Author's response to reviews

Title: Factors related to knowledge and perception of women about smoking: a cross sectional study from a developing country

Authors:

Seema Bhanji (seema.bhanji@aku.edu)
Marie Andrades (marie.andrades@aku.edu)
Fawad Taj (fawadtaj@hotmail.com)
Ali K Khuwaja (ali.khuwaja@aku.edu)

Version: 7 Date: 18 May 2011

Author's response to reviews:

May 18, 2011.

Dear Editor

BMC: Women’s Health

Subject: Responses on reviewer’s comments on manuscript 1781377658438380

Thank you very much for providing an opportunity to proof read our submitted manuscript titled: ‘Factors related to knowledge and perceptions of women about smoking: a cross-sectional study from a developing country’.

We checked the manuscript for typographic errors.

With thanks and best regards,

Corresponding author

ALI KHAN KHUWAJA
Assistant Professor and Convener Research
Family Medicine and Community Health Sciences
Aga Khan University, Karachi, Pakistan
Tel: (92-21) 486-4922
Fax: (92-21) 493-4294, 493-2095
Email: ali.khuwaja@aku.edu
Web: www.aku.edu/chs/dralikhankhowaja